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Abstract 

Benzene is a known carcinogen and can cause serious health and environmental effects. EPA 

(Environmental Protection Agency) is constantly reducing the limit of benzene content in gasoline 

due to its health effects. To meet the EPA regulations, most refiners are saturating benzene in 

gasoline to cyclohexane. Benzene saturation not only consumes useful hydrogen but also leads to 

considerable octane loss for gasoline stream. In view of this, refiners are forced to explore alternate 

ways to valorize the benzene produced in the refinery. Cyclohexylbenzene (CHB) is a versatile 

chemical with numerous industrial applications. It is used as electrolytic additive in lithium ion 

batteries, liquid crystal displays and as non-carcinogenic fuel blending additive. Furthermore CHB 

is also excellent precursor for production of petrochemical intermediates like phenol 

cyclohexanone and biphenyl. Thus, converting benzene to CHB is more profitable option in view 

of the numerous benefits of CHB.  This research is focused on development of a highly selective, 

stable and cost effective catalyst for hydroalkylation of benzene to CHB. This study also reports 

green route for the preparation of CHB avoiding the traditional two step alkylation of benzene and 

cyclohexene, using hazardous acid catalysts such as sulfuric acid and aluminium chloride. 

Theoretical understanding along with experimental studies on the effect of different physical and 

chemical properties of the catalyst on the hydroalkylation activity and selectivity is vital for 

designing efficient and cost effective hydroalkylation catalyst. Present study is attempted to 

evaluate hydroalkylation activity of different metal supported zeolites and discuss the results in 

terms of the effect of catalyst properties like zeolite acidity, zeolite pore structure, type and amount 

of metal and acid to metal ratio of the catalyst on the catalytic activity and CHB selectivity. Thus 

platinum, palladium, ruthenium and nickel modified zeolites (ZSM-5, ZSM-12, MOR, MCM-22, 

Y and Beta) were prepared and tested for hydroalkylation of benzene using fixed bed reactor at 

various operating conditions relevant to the industry. Activity of different metal loaded catalyst 

followed the sequence Ru>Pt>Pd>Ni whereas, CHB selectivity of different metal loaded catalyst 

was governed by the interaction of the reaction intermediate cyclohexene with the metal surface. 

Furthermore, hydroalkylation reaction pathway was steered by the pore geometry of the zeolites 

employed in the catalyst preparation. Among all the zeolites tested, palladium modified FAU type 

zeolite showed highest hydroalkylation selectivity. In addition to this, activity and CHB selectivity 

of the hydroalkylation catalyst were also governed by the acidity and acid/metal mole ratio of the 

catalyst. Among all the catalysts tested, Pd(0.2 wt.%)/USY12 showed highest selectivity for CHB 
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at commercially viable benzene conversion due to its optimum acidity, acid to metal mole ratio 

and uni-modal large pore system. In spite of being mono metallic catalyst with lower metal 

loading, this catalyst system gave CHB selectivity of 75% in comparison to 71% for the best 

patented/ literature reported catalysts at similar benzene conversions. Furthermore, in order to 

demonstrate the proof of concept for the conversion of benzene to CHB, kinetic modelling, process 

modelling and scale up studies were carried out on the optimum catalyst formulation employing 

in-house designed process demonstration unit. Finally, it is discussed how hydroalkylation process 

will pave way for refinery petrochemical integration assisting in the production of useful 

petrochemicals like phenolic resins, bisphenol A, caprolactam, alkylphenols, biphenyl, adipic acid, 

cyclohexanone oxime, cyclohexanone resins and nylon. 
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सार 

बेंज़ीन एक ज्ञात कैं सरकारक है और यह गंभ़ीर स्वास्थ्य और पयाावरण़ीय प्रभाव पैदा कर सकता है। ईप़ीए 

(पयाावरण संरक्षण एजेंस़ी) इसके स्वास्थ्य प्रभाव  ंके कारण गैस ल़ीन में बेंज़ीन क़ी मात्रा क़ी स़ीमा क  लगातार 

कम कर रह़ी है। ईप़ीए ननयम  ंक  पूरा करने के नलए, अनिकांश ररफाइनर गैस ल़ीन से साइक्ल एके्सन में 

बेंज़ीन संतृप्त कर रहे हैं। संतृप्तप्त बेंज़ीन न केवल उपय ग़ी हाइड्र  जन का उपभ ग करत़ी है, बप्ति गैस ल़ीन 

िारा के नलए ओकटेन क  काफ़ी  नुकसान भ़ी पहंचात़ी है। इसे देखते हए, ररफाइनर ररफाइनऱी में उत्पानदत 

बेंज़ीन क  वैि बनाने के वैकप्तिक तऱीक  ंका पता लगाने के नलए मजबूर हैं। साइक्ल हेप्तक्सलबेनज़ीन कई 

औद्य नगक अनुप्रय ग  ं के साथ एक बहमुख़ी रसायन है। इसका उपय ग नलनथयम आयन बैटऱी, नलप्तिड् 

निस्टल नड्स्प्ले में इलेक्ट्र  लाइनटक एनड्नटव और गैर-कैं सर कारक ईंिन सप्तिश्रण य जक के रूप में नकया 

जाता है। इसके अलावा साइक्ल हेप्तक्सलबेनज़ीन भ़ी नफन ल साइक्ल हेक्सन और बाइफेननल जैसे 

पेटर  केनमकल मध्यवती के उत्पादन के नलए उतृ्कष्ट अग्रदूत है। इस प्रकार, स़ीएचब़ी के कई लाभ  ंक  देखते 

हए बेंज़ीन क  स़ीएचब़ी में पररवनतात करना अनिक लाभदायक नवकि है। यह श ि स़ीएचब़ी के नलए बेंज़ीन 

के हाइड्र  केलाइजेशन के नलए एक उच्च चयनात्मक, प्तस्प्थर और लागत प्रभाव़ी उते्प्ररक के नवकास पर कें नित 

है। यह अध्ययन सल्फररक एनसड् और एलू्यम़ीननयम क्ल राइड् जैसे खतरनाक एनसड् उते्प्ररक का उपय ग 

करके बेंज़ीन और साइक्ल हेप्तक्सन के पारंपररक द  चरण क्षार से बचने के नलए स़ीएचब़ी क़ी तैयाऱी के नलए 

हरे मागा क़ी भ़ी ररप टा करता है।  हैड्र  एलकाईलेसन गनतनवनि और चयनात्मकता पर उते्प्ररक के नवनभन्न 

भौनतक और रासायननक गुण  ंके प्रभाव पर प्राय नगक अध्ययन के साथ सैद्ांनतक समझ कुशल, और लागत 

प्रभाव़ी हैड्र  एलकाईलेसन उते्प्ररक क  नड्जाइन करने के नलए महत्वपूणा है। वतामान अध्ययन में नवनभन्न िातु 

समनथात नजओलाइट्स क़ी हैड्र  एलकाईलेसन गनतनवनि का मूल्यांकन करने का प्रयास नकया गया है और 

उते्प्ररक गनतनवनि के उते्प्ररक गुण  ंजैसे नजओलाइट अम्लता, नजओलाइट निि संरचना, उते्प्ररक के िातु 

अनुपात के नलए िातु और एनसड् क़ी मात्रा और प्रकार के प्रभाव के पररणाम  ंपर चचाा क़ी गय़ी है, उते्प्ररक 

गनतनवनि पर, उदाहरण के नलए लैनटनम, पैलेनड्यम, रूथेननयम और ननकल संश नित नजओलाइट्स (ZSM-

5, ZSM-12, MOR, MCM-22, Y और Beta) तैयार नकए गए थे और उद्य ग के नलए प्रासंनगक नवनभन्न 

ऑपरेनटंग पररप्तस्प्थनतय  ंमें बेंज़ीन के हाइड्र  क्ल ररकेशन के नलए पऱीक्षण नकया गया था। नवनभन्न िातु भऱी 

हई उते्प्ररक क़ी गनतनवनि ने अनुिम Ru>Pt>Pd>Ni का अनुसरण नकया, जबनक नवनभन्न िातु ल ड् 

उते्प्ररक क़ी साइक्ल हेप्तक्सलबेनज़ीन चयनात्मकता क  िातु क़ी सतह के साथ प्रनतनिया मध्यवती 
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स्यक्ल बेंज़ीन क़ी निया से ननयंनत्रत नकया गया था। इसके अलावा, उते्प्ररक तैयाऱी में कायारत नजओलाइट्स 

के निि ज्यानमनत द्वारा हैड्र  एलकाईलेसन प्रनतनिया मागा का संचालन नकया गया था। पऱीक्षण नकए गए सभ़ी 

नजओलाइट्स के ब़ीच, पैलेनड्यम संश नित एफएयू प्रकार नजओलाइट ने सबसे अनिक हैड्र  एलकाईलेसन 

चयनात्मकता नदखाई। इसके अलावा, हैड्र  एलकाईलेसन उते्प्ररक क़ी गनतनवनि और स़ीएचब़ी चयनात्मकता 

भ़ी उते्प्ररक क़ी अम्लता और एनसड् / िातु के म ल अनुपात द्वारा शानसत ह त़ी थ़ी। पऱीक्षण नकए गए सभ़ी 

उते्प्ररक  ंमें, Pd (0.2 भार.%) / USY12 ने साइक्ल हेप्तक्सलबेनज़ीन के नलए व्यावसानयक रूप से व्यवहाया 

बेंज़ीन रूपांतरण में इसक़ी अनिकतम अम्लता, अम्ल से िातु म ल अनुपात और यूऩी-म ड्ल बडे निि 

संरचना नसस्टम के कारण उच्चतम चयनात्मकता नदखाई। कम िातु ल नडं्ग के साथ म न  मेटानलक उते्प्ररक 

ह ने के बावजूद, इस उते्प्ररक प्रणाल़ी ने समान बेंज़ीन रूपांतरण  ंमें सवाशे्रष्ठ पेटेंट / सानहत्य ररप टेड् उते्प्ररक 

के नलए 71% क़ी तुलना में 75% क़ी साइक्ल हेप्तक्सलबेनज़ीन चयनात्मकता द़ी। इसके अलावा, स़ीएचब़ी के 

नलए बेंज़ीन के रूपांतरण के नलए अविारणा के प्रमाण क  प्रदनशात करने के नलए, गनतज मॉड्नलंग, प्र सेस 

मॉड्नलंग और से्कल अप स्टड़्ीज क  इन-हाउस नड्जाइन प्रनिया प्रदशान इकाई में काम करने वाले इष्टतम 

उते्प्ररक तैयार करने पर नकया गया था। अंत में, यह चचाा क़ी गई है नक कैसे फेन नलक रेनजन, नबसफेनॉल 

ए, कैप्र लैक्ट्म, अिाइलेफेन ल्स, बाइफेननल, एनड्नपक एनसड्, साइक्ल हेके्सनॉन ऑप्तक्सन, साइक्ल हेक्सन 

रेनजन जैसे उपय ग़ी पेटर  केनमकल्स के उत्पादन में सहायता करने वाल़ी ररफाइनऱी पेटर  केनमकल एक़ीकरण 

के नलए हैड्र  एलकाईलेसन प्रनिया मागा प्रशस्त करेग़ी। 
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